Author’s response to reviews

Title: Pedi-IKDC or KOOS-Child: which questionnaire should be used in children with knee disorders?

Authors:

Charlotte van der Velden (charlottevandervelden@hotmail.com)

Marieke van der Steen (marieke.vd.steen@catharinaziekenhuis.nl)

Jens Leenders (jensleenders@hotmail.com)

Florens van Douveren (f.vandouveren@mmc.nl)

Rob Janssen (r.janssen@mmc.nl)

Max Reijman (m.reijman@erasmusmc.nl)

Version: 3 Date: 26 Apr 2019

Author’s response to reviews:

1. As you stated in your response that your study was not registered in a clinical trials registry, please remove the Trial registration section below the Abstract, as this section should only include registration numbers from trial registries.

> this has been removed from the abstract

2. In the Ethics approval section, please name the institution with which the METC is affiliated.

> the institution has been mentioned in the ethics section